Author's response to reviews

Title: The effect of long term combined yoga practice on the basal metabolic rate of healthy Adults

Authors:

M S Chaya (chayapu@hotmail.com)
Kurpad A V (a.kurpad@iphcr.res.in)
Nagendra H R (hrnagendra@rediffmail.com)
Nagarathna R (rn44@rediffmail.com)

Version: 5 Date: 2 August 2006

Author's response to reviews: see over
Response to reviewers. (Second revision).
Reviewer: Nicola Maffulli

1. OK, we have now removed the secondary aim of the paper to confirm the predictive equations. Therefore, the statements relating to this aim have been removed from the abstract, introduction as well as from results. A statement has been left to simply state that the measured BMR was comparable (or not) to the predicted BMR.

2. Reviewer: Andrew Vickers

We have diluted down the discussion to explicitly state that we cannot ascribe cause and effect due to any physiological mechanism since we did not measure them. Please see page 9: it details the caveats for this study and we have added in another stating that mechanisms cannot be readily ascribed to any specific physiological means.